Author’s response to reviews

Title: CHA2DS2-VASC score as a preprocedural predictors of contrast-induced nephropathy among patients with chronic total occlusion undergoing percutaneous coronary intervention: a single center experence

Authors:

Yong Wang (drwangyong2016@163.com)
Hong-wei Zhao (527732608@qq.com)
Xiao-jiao Zhang (434990871@qq.com)
bao-jun Chen (18940221007@163.com)
Guo-ning Yu (drwangyong321@163.com)
Ai-jie Hou (houaijie321@163.com)
Bo Luan (luanbo2016@163.com)

Version: 1 Date: 29 May 2018

Author’s response to reviews:

Dear editor :

Thank you for your careful and professional revision of this manuscript. I have revised it according your suggestion.

1. “Not applicable”has been revised in the section of Consent for publication .

2. The datasets generated and analysed during the current study are not publicly available due to a further study of this area but are available from the corresponding author on reasonable request.

3.In the Authors' contributions section, I had refered to all authors by their initials instead of their full names.

4.I had uploaded my manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.